Author’s response to reviews

Title: DNA Methylation Alterations in Grade II- and Anaplastic Pleomorphic Xanthoastrocytoma

Authors:

Ramón Martínez (ramon.martinez@med.uni-goettingen.de)
Francisco Javier Carmona Sanz (carmonasfj@gmail.com)
Miguel Vizoso (mvizoso@idibell.cat)
Veit Rohde (nchisekr@med.uni-goettingen.de)
Matthias Kirsch (matthias.kirsch@zfn.uni-freiburg.de)
Gabriele Schackert (gabriele.schackert@uniklinikum-dresden.de)
Santiago Ropero (santiago.ropero@uah.es)
Werner Paulus (werner.paulus@uni-muenster.de)
Alonso Barrantes (alonso.barrantes@med.uni-goettingen.de)
Antonio Gomez (agomezm@idibell.cat)
Manel Esteller (mesteller@idibell.cat)

Version: 6 Date: 12 February 2014

Author’s response to reviews: see over
Re: Follow-up on decision of MS: 1770274471113469 "DNA Methylation Alterations in Grade II- and Anaplastic Pleomorphic Xanthoastrocytoma"

To:

Prof Stefan Imreh and Ms Roselyn Remoto.
BMC Cancer

We are pleased to resubmit a revised version of the manuscript "DNA Methylation Alterations in Grade II- and Anaplastic Pleomorphic Xanthoastrocytoma", which has been kindly accepted for publication at BMC Cancer as a Research Article.

As requested, we have included an Author's contribution section placed after the Conflict of interest paragraph, and followed by the Acknowledgements section.

We are very grateful that you and the reviewers considered our research worthy to be published at BMC Journal.

Sincerely yours,

PD Dr. med. R. Martinez-Olivera